**Author’s response to reviews**

**Title:** A rare case of pediatric recurrent rhabdomyolysis with compound heterogenous variants in the LPIN1

**Authors:**

Ruochen Che (cheruochen@163.com)

Chunli Wang (chunliwang1@126.com)

Bixia Zheng (zbx_88117@163.com)

Xuejuan Zhang (15895908376@163.com)

Guixia Ding (bhgyuan@163.com)

Fei Zhao (zhaofei@njmu.edu.cn)

Zhanjun Jia (jiazj72@hotmail.com)

Aihua Zhang (bszah@163.com)

Songming Huang (smhuang@njmu.edu.cn)

Quancheng Feng (williamvon@126.com)

**Version:** 2  **Date:** 06 May 2020

**Author’s response to reviews:**

Dear editor and reviewers:

Thank you for considering the publication of our manuscript. The responses to the minor revision are as follows:

Editor comments
1. The table legend has been added to the main text in the end of case presentation.
2. The initials XZ in the Authors’ Contributions section has been revised. Additionally, we revised the funding information. Because firstly we plan to list Dr. Songming Huang and me as co-corresponding authors, which was not allowed by the journal. Therefore, we need to change the grant ID from Dr. Songming Huang to me. Both fundings have contributed to the manuscript.

In the decision letter, there was a paragraph for reviewer 2 report. It seems to have no relationship with our revision. I’m a little confusing, if you have any further questions, do not hesitate to let me know.

Best

Quancheng Feng
Children’s Hospital of Nanjing Medical University